Author's response to reviews

Title: Association of Socio-Economic Status with Diabetes Prevalence and Utilization of Diabetes Care

Authors:

Doreen M Rabi (doreen.rabi@calgaryhealthregion.ca)
Alun L Edwards (aedwards@ucalgary.ca)
Danielle A Southern (dasouthe@ucalgary.ca)
Lawrence W Svenson (larry.svenson@gov.ab.ca)
Peter M Sargious (peter.sargious@calgaryhealthregion.ca)
Peter Norton (norton@ucalgary.ca)
Erik T Larsen (erik.larsen@cls.ab.ca)
William A Ghali (wghali@ucalgary.ca)

Version: Date: 30 August 2006

Author's response to reviews:

Dear Editors-

Thank you for your timely response to our revised manuscript. I have reviewed the formatting changes you requested (see below) and have complied with them all.

Formatting changes requested

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT. This has been done.

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the title page. These have been removed.

Affiliations - Each affiliation should be written in full, this may mean repeating institutional addresses if the authors are from different departments within an institution, and be provided in the following format: department, institute, city, country. The affiliations are now as you have specified.

Running title - Please remove the running title from the title page. This has been removed.

Word count Please remove the word count from the title page. This has been removed.

Complete Author information - Please remove this as all the necessary information has been represented within the title page correctly. This has been removed.

Keywords section - Please remove the keywords section from the manuscript. This has been removed.

Bold text - Throughout the manuscript there are sporadic incidences of bold text. Is this for emphasis or a mistake. Please check and amend if necessary. The bolding was to illustrate where we had made our first set of revisions. We have now removed all but the intentional bolding.

Authors’ contributions - For the Authors’ contributions we suggest the following kind of format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript. Please place this section before the Acknowledgements. This has been done.

Acknowledgements - Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Our funding sources have been acknowledged.
Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

The funding information contained in Competing interests should be moved to the Acknowledgements. The Acknowledgements should be placed before the References. This has been done.

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used. This has been corrected.

Tables and figures - Please delete this title. This has been removed.

Figure title - The image file should not include the title (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded. This title has been removed.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here. I have tried my best to eliminate the white space.

Many thanks for your timely care of our manuscript. Please let me know if you need anything further.

Best regards- Doreen Rabi.